Author’s response to reviews

Title: A donor splice site mutation in CISD2 generates multiple truncated, non-functional isoforms in Wolfram Syndrome type 2 patients

Authors:
Monica Cattaneo (monica.cattaneo@multimedica.it)
Lucia La Sala (lucia.lasala@multimedica.it)
Maurizio Rondinelli (maurizio.rondinelli@multimedica.it)
Edoardo Errichiello (edoardo.errichiello01@universitadipavia.it)
Orsetta Zuffardi (orsetta.zuffardi@unipv.it)
Annibale Puca (annibale.puca@multimedica.it)
stefano Genovese (stefano.genovese@multimedica.it)
antonio Ceriello (acerbll@clinic.cat)

Version: 1 Date: 27 Jun 2017

Author’s response to reviews:
Here, the answers to the Editor's comments:

1. In the "consent to publish" was included the following sentence: Consent was obtained from all participants for publication of this result and the accompanying images. The patients provided the permission to publish the genetic data obtained from the studies performed on their blood samples.
2. Methods section was move after the background.
3. Methods were included in the abstract.
4. LC is LL. LC was changed in LL
5. More justification for the contributions of LL, MR, EE and OZ were provided.
6 Additional files was added after figure legends